Author’s response to reviews

Title: The efficacy of tumor debulking surgery is improved by adjuvant immunotherapy using imiquimod and anti-CD40

Authors:

Andrea Khong (andrea.khong@uwa.edu.au)
Amanda L Cleaver (amanda.cleaver@uwa.edu.au)
Muhammad Fahmi (fahmi.alatasdr@gmail.com)
Ben C Wylie (bwylie@ichr.uwa.edu.au)
Theresa Connor (tconnor@ichr.uwa.edu.au)
Scott A Fisher (Scott.fisher@uwa.edu.au)
Steve Broomfield (S.Broomfield@murdoch.edu.au)
Willem J Lesterhuis (willem.lesterhuis@uwa.edu.au)
Andrew J Currie (A.Currie@murdoch.edu.au)
Richard A Lake (Richard.lake@uwa.edu.au)
Bruce W Robinson (bruce.robinson@uwa.edu.au)

Version: 2
Date: 11 September 2014

Author’s response to reviews:

Dear Editor,

The following changes have been made to the manuscript in response to the initial review made by the editorial office:

1. All font changed to Times New Roman, size 12 pt, 2 line spacing
2. Line numbers added
3. All page breaks removed
4. Title page includes details of submitting author (Line 18)
5. Details for co-authors are included (Line 25-35)

Kind regards,

Andrea Khong